Author's response to reviews

Title: Clinical effectiveness of post-operative splinting after surgical release of Dupuytren's contracture: a systematic review

Authors:
  Debbie Larson (debbie.larsen@nnuh.nhs.uk)
  Christina Jerosch-Herold (c.jerosch-herold@uea.ac.uk)

Version: 3 Date: 21 July 2008

Author's response to reviews: see over
The following requested changes have been made:

1. page numbers have been removed  
2. in abstract - ‘Methods/design’ has been changed to ‘Methods’  
3. in abstract - ‘Discussion’ has been changed to ‘Conclusions’  
4. Acknowledgements header removed  
5. ‘Figures’ has been changed to ‘Figure legends’  
6. Tables have been renamed ‘additional files’  
7. Table legends section has been removed and is now included in the additional files